Author's response to reviews

Title: Economic evaluation of the practical approach to lung health and informal provider interventions for improving the detection of tuberculosis and chronic airways disease at primary care level in Malawi: Study protocol for cost-effectiveness analysis.

Authors:

Elvis S Gama (elvis.gama@lstmed.ac.uk)
Jason J Madan (j.j.madan@warwick.ac.uk)
Hastings M Banda (hastings@reachtrust.org)
Bertie Squire (bertie.squire@lstmed.ac.uk)
Rachael Thomson (r.thomson@lstmed.ac.uk)
Ireen Namakhoma (ireen@reachtrust.org)

Version: 2
Date: 11 December 2014

Author's response to reviews:

The Editor
Implementation Science
11th December 2014

Dear Editor,

REVISION OF MANUSCRIPT

I am writing to resubmit our manuscript entitled, “Economic evaluation of the practical approach to lung health and informal provider interventions for improving the detection of tuberculosis and chronic airways disease at primary care level in Malawi: Study protocol for cost-effectiveness analysis”, for consideration for publication in Implementation Science.

Based on the advice provided in your email, the following change has been made;

1. The methods section has been revised to follow the CHEERS standard.

Thank you for receiving our manuscript and considering it for publication. We appreciate your time and look forward to your response.

Yours Sincerely

Elvis Gama, PhD
Liverpool School of Tropical Medicine
Pembroke Place
L3 5QA
Liverpool
elvis.gama@lstmed.ac.uk

Enclosure: Protocol Manuscript and supporting documents